Psychology of Sexual Orientation and Gender Diversity

Understanding the Experiences of Youth Who Have Discontinued a Gender Transition: Provider Perspectives

Annie Pullen Sansfaçon, Tommly Planchat, Morgane A. Gelly, Alexandre Baril, Françoise Susset, and Mélanie Millette Online First Publication, May 11, 2023. https://dx.doi.org/10.1037/sgd0000644

CITATION

Pullen Sansfaçon, A., Planchat, T., Gelly, M. A., Baril, A., Susset, F., & Millette, M. (2023, May 11). Understanding the Experiences of Youth Who Have Discontinued a Gender Transition: Provider Perspectives. *Psychology of Sexual Orientation and Gender Diversity*. Advance online publication. https://dx.doi.org/10.1037/sgd0000644



© 2023 American Psychological Association

https://doi.org/10.1037/sgd0000644

Understanding the Experiences of Youth Who Have Discontinued a Gender Transition: Provider Perspectives

Annie Pullen Sansfaçon^{1, 2}, Tommly Planchat¹, Morgane A. Gelly¹, Alexandre Baril^{2, 3},

Françoise Susset⁴, and Mélanie Millette⁵

¹ School of Social Work and Canada Research Chair on Transgender Children and Their Families and School of Social Work, Université de Montréal

² School of Social Work, University of Stellenbosch

³ School of Social Work, University of Ottawa
⁴ Centre de santé Méraki, Montreal, Quebec, Canada

⁵ Departement of Communication, Université du Québec à Montréal

The idea of "detransition" is stimulating debates among researchers and providers working with trans and nonbinary youth (TNBY), but research on this phenomenon is still in its early stages. This study presents the characteristics and experiences of providers who work directly with TNBY, including some who have discontinued a transition. Sixty-one providers in trans health care were recruited internationally and responded to an online survey. They come from various regions of the world and disciplines and 77% worked according to the trans-affirming approach. Sixty-four percent of them had followed a youth who discontinued their transition. Among those, 82% reported a range of 1–5 youth in their entire career. Professionals reported their observations of discontinuation with regard to the youth's individual characteristics, and parental and social support. The results suggest that professionals had experience with youth who had discontinued regardless of their approach to intervention (e.g., trans-affirming, watchful waiting, or exploratory) and have observed a diversity of characteristics when describing the youth they followed.

Public Significance Statement

We found that professionals who have followed trans and nonbinary youth (TNBY) who have since discontinued their gender transition practice according to a range of intervention approaches (trans-affirming, wait and see, or other approaches) and that more than half have over 5 years of experience in trans health care. Observations from their practice show that youth present varied individual characteristics, as well as levels of parental and social support, pointing toward a diversity of experiences and pathways among youth who were followed by the providers. Our results suggest that future research should examine how to best support youth, whatever their gender journey or the outcome, rather than preventing it.

Keywords: detransition, discontinuation of transition, trans and nonbinary youth, trans care providers, youth characteristics

Over the last few years, clinics in the United States have noted an increase in trans and nonbinary youth (TNBY)¹ seeking gender-affirming care (Respaut & Terhune, 2022). This is despite recent evidence showing that access to gender-affirming care remains challenging (Ashley, 2019a; Gridley et al., 2016; Puckett et al., 2018) and that wait times to be seen by a provider are long (Bauer et al., 2021). Accessing gender-affirming treatments is important as research has shown that medical gender transition can alleviate distress and improve the well-being of those who experience gender incongruence (Owen-Smith et al., 2018; Pullen Sansfaçon,

Annie Pullen Sansfaçon Dhttps://orcid.org/0000-0003-2286-8997
Correspondence concerning this article should be addressed to
Annie Pullen Sansfaçon, School of Social Work, Pavilion Lionel Groulx,
Université de Montréal, CP6128 Succ. Centre Ville, Montreal, QC H3C3J7,
Canada. Email: a.pullen.sansfacon@umontreal.ca

Medico, et al., 2023; Sorbara et al., 2020; Turban et al., 2020) and reduced gender dysphoria (van Leerdam et al., 2023). For example, after 12-month follow-up, youths' depression is reduced by 60% and suicidality by 73%; for those who did not access care, the team observed that the risk increased by 2–3 times (Tordoff et al., 2022). Access to gender-affirming care is also identified as an important factor leading to well-being in research that directly reports TNBY experiences (Pullen Sansfaçon, Medico, et al., 2023). Gender-affirming care also continues to be recommended by the World Professional Association of Transgender Health (WPATH) in their newest Standards of Care (Coleman et al., 2022).

Among TNBY who begin a transition, some will experience a shift in their gender journey (Cohen et al., 2022), or change the course of their gender transition (Durwood et al., 2022). This phenomenon is sometimes called a "detransition." This refers to

¹Our terminology includes all gender diverse and creative youth.

discontinuing a gender transition, which Vandenbussche (2022) described as complex to define and poorly understood. There is no universal definition of detransition (Hall et al., 2021) and different terminologies and definitions have been proposed as part of the detrans umbrella (Hildebrand-Chupp, 2020). For some, detransition means reversing a process of gender transition and can include social and/or medical cessation (Hildebrand-Chupp, 2020) or performing inversion treatments and/or surgery (Littman, 2021). A "detransitioner" refers to a person who embarks on this process but more importantly, identifies with the label; this may or may not include a sense of belonging to a "detrans community" (Vandenbussche, 2022). Sometimes, the term "retransition" is also used but it can also mean resuming a gender transition after having "detransitioned" (e.g., reaffirming a trans or nonbinary identity after having affirmed a cisgender identity; Hildebrand-Chupp, 2020). Another terminology that is sometimes used is "desistance," which usually refers to a renouncement of trans identification or a cessation of gender dysphoria (Karrington, 2022). "Desistance" is also often used to discuss the gender journey of preadolescent children who cease to express or identify as gender diverse (Karrington, 2022), in which case it does not involve a medical transition or a "detransition" as defined earlier. We also note that the term "discontinuation" has recently been introduced in the literature as a less stigmatizing term and is understood as the process of ceasing a transition process or care (e.g., Turban et al., 2018; Vrouenraets et al., 2022).

Some authors have hypothesized that discontinuation is a "growing phenomenon" (Marchiano, 2017) caused by an increase in the popularity and use of trans-affirming and informed consent approaches with youth, rendering access to gender-affirming care easier for them (Littman, 2021). Although there are some individuals who choose to discontinue their transition, growing evidence suggests this is infrequent in occurrence (Brik et al., 2020; Fornander, 2022; Hall et al., 2021; James et al., 2016; Olson et al., 2022; Turban et al., 2021). In a study that included older children, only a few youths discontinued blocker treatment, and 3.5% no longer wanted to receive gender-affirming treatment (Brik et al., 2020). In a retrospective study of an online U.S. survey with 17,151 people identifying as transgender (James et al., 2016), 13.1% reported having discontinued, at some point in the past but participants in that study currently identify as TNBY (Turban et al., 2021). Another retrospective study based on a review of case notes between 2017 and 2018 in a gender clinic found that 6.9% of patients discontinued and reverted to living in their original gender role (Hall et al., 2021). A recent longitudinal study of socially transitioned children found that 2.5% discontinued to return to a cisgender identity, and a further 3.5% was not living in a nonbinary identity (Olson et al., 2022). This newest research is contrasting with previous research on "desistance" (e.g., Cohen-Kettenis et al., 2011; Drummond et al., 2008; Wallien & Cohen-Kettenis, 2008), which estimated that about 80% of prepubescent gender-diverse, gender nonconforming and trans youth ceased to identify as such when reaching adolescence. It is important to note that these past studies have received substantial criticism for their methodology and their interpretation of data (see Ashley, 2019b, 2022; Karrington, 2022; Temple Newhook et al., 2018; Winters, 2017; Winters et al., 2018). Even if it appears to be infrequent, the idea of discontinuation² is beginning to raise concerns among some gender-affirming care providers and revives debates about the appropriateness of the trans-affirming approach for working with TNBY (Ashley, 2019a; de Vries & Cohen-Kettenis, 2012; Edwards-Leeper et al., 2016; Ehrensaft et al., 2018; Littman, 2021; Spiliadis, 2019; Steensma et al., 2011, 2013).

This phenomenon has also captured much media attention lately and researchers have observed a sharp increase in the publication of news articles on this topic between 2017 and 2020 (Paré-Roy et al., 2022). Media often frame the experience of young people who have discontinued a transition (YDT) as a mistake and experience inevitably leading to regret, and as an outcome resulting from a misdiagnosis by their providers (Paré-Roy et al., 2022; Slothouber, 2020). Media coverage is said to have a negative effect on access to trans health care (Indremo et al., 2022). Research on discontinuation is still in its early stages (Littman, 2021) and the experiences of YDT are still poorly understood (Expósito-Campos, 2021). Faced with various, sometimes contradictory evidence, it is therefore essential to better understand this phenomenon and gain a boarder perspective on the topic. This article examines this phenomenon from the perspective of the providers working in trans health care. To our knowledge, it is the first research to pay attention to their experience. The article presents the observations and perspectives of providers who have followed or encountered YDT in their practice. The next section of the article presents the literature on discontinuation, the reasons for individuals to discontinue their transition as well as some of the characteristics that may be present, including the possible regret involved with decisions about transitions and discontinuation, and the role providers may play in supporting those processes.

Characteristics Related to a Discontinuation of Transition

Some studies have shown that some YDT will return to a cisgender identity, while others will evolve toward a nonbinary or another identity (Cohen et al., 2022; Durwood et al., 2022). In another study, 60% of participants reported returning to a cisgender identity while others reported their identity evolving to something else (Littman, 2021). Discontinuation may only be temporary (Turban et al., 2021). Some studies have attempted to identify characteristics among YDT (Fornander, 2022; Vandenbussche, 2022).

A mixed-method study conducted in a U.S. clinic with 829 TNBY suggests that discontinuation is not always due to a change of identity but some differences related to physical and mental health can be found between those who continue and those who discontinue a gender transition (Fornander, 2022). Another study shows that 54% of YDT self-reported at least three coexisting conditions, the most prevalent ones being depression (70%), anxiety (63%), and a posttraumatic stress disorder (PTSD; 33%; Vandenbussche, 2022). However, it must be noted that mental health issues are prevalent among the TNBY population, and that those are often caused by minority stress, as well as by other social causes such as a lack of social support, nonaffirming living environments, or lack of access to mental health or gender-affirming services which all have a significant negative impact on them (Delozier et al., 2020; Jones et al., 2021; Pellicane & Ciesla, 2022). Other characteristics that were identified related to sexual orientation (being bisexual), levels of support

²While so far the terms "detransition," "desistance," and "discontinuation" have been used to convey how other authors present their work in the literature consulted, this article now use the term *discontinuation* which is encompassing of all terminologies presented.

(having a nonsupportive family), and the type of transition undertaken (not having medically transitioned; Turban et al., 2021).

Reasons Possibly Leading to a Discontinuation of Transition

The reasons stated for discontinuing a transition are varied. Some discontinue because they realize that their feeling of gender dysphoria was related to some other issues (e.g., internalized misogyny or homophobia, mental health; Littman, 2021; Vandenbussche, 2022). Health concerns, dissatisfaction with the physical results, nonalleviation of gender dysphoria, feeling more comfortable with their assigned gender at birth, or change in political views were also mentioned (Littman, 2021; Vandenbussche, 2022). External pressures (e.g., pressure from parents, family members, partner or community, social stigma, and difficulty finding work) are often cited as a source of temporary discontinuation (Turban et al., 2021).

The Question of Regret

Research publications about regret or about satisfaction with gender-affirming surgery are still rare and point toward different results, mainly because they employ different methodologies (retrospective analysis, qualitative and quantitative studies) and draw on different samples (children, TNB and YDT, TNB adults) which makes the comparison between the studies challenging. For example, some research highlights that feelings of regret are rare in the TNB population (Dhejne et al., 2014; Narayan et al., 2021; Olson-Kennedy et al., 2018; Wiepjes et al., 2018). The meta-analysis of Bustos et al. (2021) estimates the rate of regret around 1%, with slight differences between "transfeminine" surgery (CI: <1%-2%) and "transmasculine" surgery (CI: <1%-<1%). However, the feeling of regret experienced by YDT is less studied and information varies greatly depending on the studies. Some studies have found that regret is rare (Cohen et al., 2022) or nonexistent (Durwood et al., 2022) while some studies have found high levels of regret (Littman, 2021; Vandenbussche, 2022).

Other studies have shown that people can experience gratefulness for their transition journey (Turban et al., 2018; Turban & Keuroghlian, 2018). In the research of MacKinnon et al. (2021), 67% expressed no regret and/or positive feeling about their past medical interventions, and 22% mentioned regret and thought that medical transition was the wrong way for them. Another study shows that regret and gratitude can coexist, sometimes leading to ambiguous experiences (Pullen Sansfaçon, Gelly, et al., 2023)

Lack of Support While Undertaking a Discontinuation

Some studies have found that people often discontinue their medical transition abruptly without supervision (MacKinnon et al., 2022). Support is nevertheless important as YDT have important needs and may feel in distress at the time of discontinuation (Vandenbussche, 2022). YDT would tend to disengage from their follow-up, out of mistrust of providers, and turn to the internet and social networks for support (MacKinnon et al., 2022; Vandenbussche, 2022). Overall, support for YDT appears to be lacking; studies have highlighted that people who discontinued a transition and who benefited from some support (e.g., to modify or cease their hormonal treatment), had negative experiences because the

professional who provided the support lacked adequate knowledge or was being judgmental (MacKinnon et al., 2022).

Current Study

The data presented in this article was drawn from a larger mixedmethod exploratory study about the discourses surrounding the discontinuation of gender transition among youth. The larger project aimed at examining the differences between the perspectives of YDT (Study 1), the providers who work in the field of trans health (Study 2), and the discourses found in the press and on social media on YDT (Study 3). The research team is composed of cis and TNB researchers, research assistants, and a clinician as well as one YDT who acts as a consultant to the team. Some team members are members of WPATH. The project goal is to better understand discontinuation as a phenomenon in order to improve interventions with TNBY, whether they pursue or discontinue their gender transition in the future. This article is based on the data obtained in the context of Study 2 from the perspective of the providers working with TNBY. Providers were invited to complete an online survey collecting information about their practice's characteristics, and their experience with TNBY and with YDT.

Method

Procedure

Participants were invited to respond to an online survey. The survey was located on LimeSurvey and was opened from September 1, 2020 to January 31, 2021 to trans-care providers from all disciplines and working with TNBY. The survey was presented as a study that aimed to understand the perspectives and experiences of providers on discontinuation, "detransition" and "desistance" as definitions of those terms were still debated at the start of the research in 2019.

Participants were recruited through an electronic invitation circulated on various Listserv forums (e.g., WPATH, World Association of Sexology, Working Group on Gender (USA) as well as on different provider Facebook groups (e.g., International transgender health, and Professionnel.le.s en santé trans). The invitation was also shared by international and European trans-health organizations, as well as organizations working with TNB people nationally (USA, Canada, including French-speaking Quebec, Australia, France, England, Italy, and Switzerland).

On the first page of the survey, potential participants were given information about the research (e.g., purpose, involvement, eligibility criteria, confidentiality, inconvenience and benefit to participating, etc.). They had to give their consent by ticking "I agree to participate in this research project." The research project received ethical approbation from the ethics board of the principal investigator at the University of Montreal, as well as from the coinvestigators' institutions: Université du Québec à Montréal, and the University of Ottawa.

Measures

The questions in the survey were developed by the research team based on the knowledge of discontinuation at the time. At the beginning of the project, empirical evidence on discontinuation was still very scarce, so the team drew on the literature on desistance and on the idea of "rapid onset gender dysphoria," two ideas that were circulating widely at the time, as well on their clinical experience and on empirical evidence and approaches to working with TNBY (e.g., gender affirmation, wait and see, and corrective approaches). The tool was not validated before the survey. The main sections of the survey are detailed below.

Provider Characteristics

The survey presented a series of questions about their practice location and context and the approaches used with TNBY under 25 years old (e.g., "Upon which intervention approach do you mainly base your professional practice with respect to trans youth?" with multiple-choice options: Trans-affirming approach; Watchful waiting approach [Amsterdam approach]; Corrective approach; Other approaches, specify). Participants were also asked about whether they used this approach with prepubescent or preteen youth (multiple-choice options: Yes; no; not applicable). Their experience with TNBY was obtained by asking a close-ended question about the total number of youth they had followed in their career using the ranges provided (1-10 youth; 11-25 youth; 26-50 youth; 51 youth, or more). For some questions, participants were asked to specify their answer, for example, in one closed-ended question, "Do you have specialized training in trans health care?" (No; Yes, [please specify]).

Attitudes and Beliefs About Gender Transition

All the participants were asked to respond to a series of 18 statements about principles that guided their practice with TNBY. They were asked to indicate their adhesion to statements such as "An adolescent who identifies as nonbinary should not start gender-affirming hormone treatments," "there are people who feel gender neutral, neither man nor women," and "parental behavior often leads the youth to identify as trans" and provide their answer on a Likert scale (strongly agree; agree; uncertain; disagree). Participants were also asked if they practiced according to a model of informed consent (answer: no; yes) and who makes the final decision with respect to the intervention offered (answers: me, the professional; the young person/patient; Me, the professional, in collaboration with the person/patient; other, specify).

Discontinuation of Transitions

The survey asked participants about whether or not they had followed YDT with three choices of answers (yes, no, maybe). Those who choose "yes" or "maybe" options had to identify the total of youth they followed with a range scale (1-5 youth; 6-10 youth; 11-20 youth; 21 youth, and more). Participants were asked to describe what they observed about the most recent situation of discontinuation with a series of 36 statements (individual characteristics of the youth, parental and family support, and social support-e.g., "The youth returned to an identity linked to the sex/gender they were assigned at birth."; "the parents reported several concerns in response to the transition"; "the youth had access to the psychosocial services they needed"). Participants were offered choices on a Likert scale (yes: describes the situation well; a bit: somewhat describes the situation; no: does not describe the situation; not applicable/I don't know). This section also includes one open-ended question asking participants to describe a typical situation of discontinuation, and one open-ended question asking them to describe what they thought were the reasons for discontinuation.

Belief About Discontinuation

The final section included 19 statements examining their beliefs about discontinuation (e.g., "all I can do is accompany a youth in their experience"; "I think it is possible to identify a young person who will not follow through with transition later on.") answers were provided on a Likert scale (strongly agree; agree; uncertain; disagree). The survey was concluded by three open-ended questions asking participants to identify which authors or texts influence the most their practice with TNBY, what should be paramount when working with TNBY, and any other opinion they wish to share on discontinuation of transition.

Data Analysis

Descriptive statistical analysis using SPSS version 26 was used to examine providers' characteristics and practice, as well as their experience with discontinuation. Data was first extracted from LimeSurvey servers. Some qualitative measures were categorized, for example, for location, the names of the cities or regions were generalized by the name of the country. We created a coding for the nominal categories so that they could be processed in the SPSS software. Once all the data were coded, a descriptive statistical analysis was performed with SPSS software to obtain the frequencies, percentages, and mode of each response.

Description of Sample

A total of 147 participants took part in the online survey, and 61 of them completed more than 60% of the survey including the section on discontinuation. The following data only describe the characteristic of the participants (Table 1).

The participants answered in English (64%) and in French (36%). Participants mainly practiced in North America, the United States (29.5%) and Canada (27.9%), or Europe, mostly Switzerland (19.7%) and the United Kingdom (8.2%). They worked largely (63.9%) in the disciplines of psychology or medicine, and some in social work (11.5%). Other disciplines of work included youth intervention, sexology, counseling, surgery, speech therapists, gynecology, etc. Most of the providers worked according to gender-affirming approach (77%), but 6.6% of them worked according to the Dutch "wait and see" approach. No one affirmed working according to the corrective approach but 16.4% of participants selected "other" for the intervention approach, providers mentioned working according to psychotherapeutic exploration, mixed approaches with watchful waiting, regular therapy, thorough evaluation, etc.

Results

Experience With YDT

Over half of the sample (56%) reported having followed YDT and 8% of participants thought they had maybe followed some. The remaining sample (36%) did not report any case of YDT in their entire career.

Among participants who had or may have followed YDT (n = 39), most of them (82%) reported a range between 1 and 5 YDT in their whole career. The other who had followed YDT, 8% reported a range between six and 10 youth, 2% reported between 11 and 20 youth, and 8% reported having followed 21 youth and

 Table 1

 Providers' Practice Characteristics and Experience

Baseline measure	% (n)	N = 61
Geographical region		
North America	57.4 (35)	
Europe	39.3 (24)	
Oceania	3.3 (2)	
Discipline		
Psychology	39.3 (24)	
Medicine	24.6 (15)	
Social work	11.5 (7)	
Other	24.6 (15)	
Practice setting		
Private	37.7 (23)	
Public	37.7 (23)	
Community organization	19.7 (12)	
Other	4.9 (3)	
Years of experience with TNBY	` '	
≤2	8.2 (5)	
- 2-5	37.7 (23)	
5–10	29.5 (18)	
≥10	24.6 (15)	
Total of TNBY served in their career		
1–10 youth	11.5 (7)	
11–25 youth	26.2 (16)	
26–50 youth	18.0 (11)	
≥51 youth	44.3 (27)	
Special training in trans health care		
Yes	72.1 (44)	
No	27.9 (17)	
Intervention approaches		
Gender-affirming	77.0 (47)	
Watchful waiting (Amsterdam)	6.6 (4)	
Corrective	0.0(0)	
Other	16.4 (10)	
Similar approach with prepubescent TNBY	. ,	
Yes	67.2 (41)	
No	3.3 (2)	
N/A	29.5 (18)	
Practice model of informed consent	. ,	
Yes	78.7 (48)	
No	21.3 (13)	

Note. TNBY = trans and nonbinary youth.

more. Most providers who reported six or more YDT had worked with 51 or more TNBY in their entire careers. These providers were mostly located in Europe and working with another intervention approach (e.g., psychotherapeutic exploration, regular therapy, thorough assessment, etc.). We have observed that participants practicing all sorts of intervention approaches reported cases of YDT in their practice (Table 2).

Providers' Observations on YDT Characteristic

When participants were asked to think about the most recent situation of discontinuation in their practice, 40% of the providers observed that YDT returned to an identity congruent to their gender assigned at birth, 31.4% reported that the YDT reidentified with the gender assigned at birth a bit and 25.7% described that the YDT did not reidentify to their gender assigned at birth. Over half of the participants (52.9%) did not report an expression of regret from the YDT, but 20.6% of the providers reported that YDT had expressed regret, and 20.6% had perceived a bit of regret in the situation. A larger proportion of participants (58.8%) stated that the YDT had experienced mental

health problems before starting the transition. As for the other participants, 23.5% considered this to be somewhat descriptive of the YDT situation, and 17.6% did not observe any mental health issues before transitioning (Table 3).

Over a third of the providers (35.3%) described youth as being well supported by their families as well as by providers. We found the same results regarding the improvement of the mental health of the youth since the beginning of the transition: 35.3% of the participants saw improvement, and 35.3% did not. Some providers (38.2%) said the youth did not seem happy to have started, and then stopped their transition while another 35.5% reported youth being happy to have done it.

Providers' Observations of Parental and Family Support

In their most recent situation of discontinuation encountered in their practice, half of the providers described that the parents took the time to accept and support the youth, 26.5% said the parent accepted a bit, and 20.6% did not at all. For 41.2% of the participants, the parents had no doubts about their child's trans identity, 29.4% reported that the parents had doubts about it, and 23.5% mentioned that this somewhat described the situation. Part of the providers (41.2%) reported that parents were not reluctant to transition, 32.4% reported that parents were reluctant, and 17.6% of the providers, this described the situation a bit. For most of the participants (67.7%), the family did not appear to have had any pressure from their close circle not to allow their children to transition. Also, a large proportion of the providers (61.8%) reported that no one in the family seemed affected negatively by the coming out of the youth. However, 20.6% said they did not know that information (Table 4).

Providers' Observations of Youth's Social Support

In the most recent situation of discontinuation, most providers (70.6%) observed that the youth seemed well supported by their friends during their transition. Other providers (11.8%) did not know if it was the case, 14.7% said the youth was a bit supported, and only 2.9% reported the youth was not supported. Most participants (67.5%) reported that the youth had access to medical services they needed, 14.7% a bit, and only 2.9% did not have access. Over half of the participants (58.8%) observed that the youth had access to psychosocial services. Other providers (29.4%) reported that they had some access to psychosocial services, and 8.8% stated that the youth had no access. A large portion of the sample (64.7%) said that the youth did not feel pressure from their friends about the way to live their gender. Other participants (23.5%) reported a bit of pressure, and 5.9% reported pressure. Half of the participants (50%) reported that the youth who discontinue had access to a supportive community, 35.3% a bit, and 8.8% had no access to it. Half of the providers (50%) described the school as supportive during the youth's transition, whereas 23.5% reported a bit, 17.6% did not know, and 8.8% described the school as nonsupportive (Table 5).

Discussion

The purpose of this article was to examine providers' characteristics and their experience with YDT. Providers have observed a diversity of characteristics about YDT, including 36% who never met any in all their careers. Among the 64% who did, the majority (82%)

 Table 2

 Providers' Experience With YDT and Approaches Used in Practice

Have met / number of youth	Gender- affirming % (n)	Dutch approach % (n)	Other approach % (n)	Total % (n)	N = 61
Do you had follow TNBY who	discontinue?				
Yes	36.8 (22)	5 (3)	14.7 (9)	55.8 (34)	
Maybe	8.1 (5)	0 (0)	0 (0)	8.2 (5)	
No	32.8 (20)	1.6(1)	1.6(1)	36 (22)	
Number of YDT					
0 youth	32.8 (20)	1.6(1)	1.6(1)	36 (22)	
1–5 youth	42.6 (26)	1.6(1)	8.2 (5)	52.4 (32)	
6–10 youth	1.6(1)	0 (0)	3.2(2)	5 (3)	
11–21 youth	0 (0)	0 (0)	1.6(1)	1.6(1)	
≥21 youth	0 (0)	3.2 (2)	1.6 (1)	5 (3)	

Note. TNBY = trans and nonbinary youth; YDT = young people who have discontinued a transition.

reported only following a total of one to five cases. This seems to echo recent studies which suggest that discontinuation is infrequent (Brik et al., 2020; Fornander, 2022; Hall et al., 2021; James et al., 2016; Olson et al., 2022). However, other studies show that a large proportion of people who discontinue do not inform their clinician or clinic (Littman, 2021; MacKinnon et al., 2022) or experienced a service breakdown (MacKinnon et al., 2022; Vandenbussche, 2022). Practitioners should therefore remain aware of the possibility that youth who thinks about discontinuing may not feel well supported, or confident about talking to their providers about that. This may be facilitated by providers making it clear from the beginning that their support is unconditional and that there is no shame in readjusting their transition pathways or rediscussing intervention needs.

In our study, the providers had experience with youth who discontinue regardless of their approach to intervention (trans-affirming, watchful waiting, exploratory, etc.). This is at odds with Littman's hypothesis (2021) that youth are at increased risk of discontinuation when they are followed in trans-affirming approaches rather than

other approaches, including exploratory approaches. Our results also show that providers who practice a wait-and-see approach also see youth who discontinue their transition (Expósito-Campos, 2021). Hence, our results about the observations of youth discontinuation by providers show no evidence that the approach is correlated with lower rates of regret or discontinuation.

The providers have observed varied characteristics among YDT they met. As in Littman's (2021) study, trans health care providers in our research observed a range of identifications, including cisgender and other identities. Recent studies show that the gender identity of young people may evolve from a trans binary to a nonbinary or cisgender identity several times over time (Olson et al., 2022; Pullen Sansfaçon, Gelly, et al., 2023). That said, the discontinuation of transition among TNBY did not always involve a change or fluctuation in gender identity (Fornander, 2022). Hence, clinicians should promote flexibility in gender identity and support individuals in their self-identification in the present moment, whether this constitutes a continuation or discontinuation of the previously affirmed gender identity (Baril & Silverman, 2019).

Table 3Providers' Observations on YDT Characteristic

Item	Yes % (n)	A bit $\%$ (n)	No % (n)		N = 35
The youth returned to identify at sex/gender birth	40 (14) ^a	31.4 (11)	25.7 (9)	+1	
The youth showed gender fluidity from the beginning	28.6 (10)	17.6 (6)	$54.3 (19)^{a}$		
The youth had mixed emotions with the perception of others ^b	26.5 (9)	$38.2 (13)^{a}$	29.4 (10)	+2	
The youth spent time solely with TNB people before transition ^b	26.5 (9)	14.7 (5)	$52.9 (18)^{a}$	+2	
The mental health of the youth has improved since the beginning of the transition ^b	$35.3(12)^a$	26.5 (9)	35.3 (12) ^a	+1	
The youth experienced mental health issues before beginning transition ^b	58.8 (20) ^a	23.5 (8)	14.7 (5)	+1	
After undergoing medical treatment, the youth realized they did not want to do a binary transition ^b	29.4 (10)	$35.3(12)^a$	29.4 (10)	+2	
The youth expressed doubt and decided to postpone medical treatment ^b	14.7 (5)	23.5 (8)	52.9 (18) ^a	+3	
The youth expressed regrets at having started gender transition ^b	20.6 (7)	20.6 (7)	52.9 (18) ^a	+2	
The youth discontinue as the treatment undergone up to this point had allowed them to achieve well-being					
and pursuit no longer seemed necessary ^b	17.6 (6)	23.5 (8)	$58.8 (20)^{a}$		
The youth felt mixed emotions with respect to physical changes ^b	17.6 (6)	32.4 (11)	$41.2 (14)^{a}$	+3	
The youth seemed happy to have begun and then stopped their transition ^b	35.3 (12)	17.6 (6)	38.2 (13) ^a	+3	
During the process, the youth seemed capable of undertaking a process without external constraints ^b	50 (17) ^a	38.2 (13)	11.8 (4)		
The youth described being very well supported by their parents during their transition ^b	$35.3(12)^a$	26.5 (9)	35.3 (12) ^a	+1	
The youth reported feeling a certain amount of pressure from their parents ^b	14.7 (5)	32.4 (11)	$50(17)^{a}$	+1	

Note. Response options: Yes: describes the situation well; A bit: somewhat describes the situation; No: does not describe the situation; N/A: not applicable/I do not know; YDT = young people who have discontinued a transition. a Mode, the most frequently represented value. ${}^{b}n = 34$ for these items. + Number of participants who responded N/A.

 Table 4

 Providers' Observations on YDT'Parental and Family Support

Item	Yes % (n)	A bit % (n)	No % (n)		N = 34
I felt reticence on the part of at least one of the parents concerning the taking of medicine(s)	35.3 (12) ^a	20.6 (7)	32.4 (11)	+4	
The parents took the time to accept their child and support the transition	$50(17)^{a}$	26.5 (9)	20.6 (7)	+1	
The parents doubted that the youth was trans	29.4 (10)	23.5 (8)	$41.2 (14)^{a}$	+2	
The parents respected the youth in the use of their chosen name	$61.8 (21)^a$	14.7 (5)	20.6 (7)	+1	
The parents always respected the pronouns chosen by the child	$47.1 (16)^{a}$	23.5 (8)	26.5 (9)	+1	
The parents frequently used the wrong pronouns	17.6 (6)	26.5 (9)	$47.1 (16)^{a}$	+3	
The parents reported several concerns in response to the transition	$41.2 (14)^{a}$	23.5 (8)	29.4 (10)	+2	
I felt reticence on the part of the parents with respect to the transition	32.4 (11)	17.6 (6)	$41.2 (14)^{a}$	+3	
The parents were reluctant to participate in a support group	26.5 (9)	14.7 (5)	$29.4 (10)^{a}$	$+10^{a}$	
The parents aligned easily with a professional opinion	$35.3(12)^a$	23.5 (8)	29.4 (10)	+4	
The parents sought support from other parents supporting a trans youth	17.6 (6)	14.7 (5)	$41.1 (15)^{a}$	+8	
The family felt pressure from their circle not to allow the youth to transition	8.8 (3)	11.8 (4)	$67.6 (23)^{a}$	+4	
Someone in the family seemed negatively affected by the coming-out	2.9(1)	14.7 (5)	$61.8(21)^{a}$	+7	
Someone in the family was violent with the youth throughout their coming out	5.9 (2)	8.8 (3)	$73.5(25)^{a}$	+4	

Note. Response options: Yes: describes the situation well; A bit: somewhat describes the situation; No: does not describe the situation; N/A: not applicable/I don't know; YDT = young people who have discontinued a transition. ^aMode, the most frequently represented value. + Number of participants who responded N/A

Our results have shown that more than half of the providers (58%) observed that youth had a mental health issue before they began to transition. The presence of mental health issues at the beginning of the transition was also found in other studies about discontinuation (Fornander, 2022; Littman, 2021; Vandenbussche, 2022). Studies about TNBY show that they often experience mental health issues such as depression when they first embark on medical gender-affirming care, but that symptoms tend to improve after the start of the treatment (Pullen Sansfaçon et al., 2019; Tordoff et al., 2022).

Research studies on "discontinuers," whether on adults or on YDT also discuss the presence of mental health issues in these groups, pointing to complex mental health issues that may be related to trauma as well as the fact that a lot of "discontinuers" are still a part of the sexual and gender minority group of people (Littman, 2021; Turban et al., 2021; Vandenbussche, 2022). Research studies show that sexual and gender minority groups experience high levels of minority stress which impacts their physical and mental health (Hunter et al., 2021; Meyer, 2003; Rood et al., 2016). Gender dysphoria may also still be an issue for many people who discontinue (MacKinnon et al., 2022; Pullen Sansfaçon, Gelly, et al., 2023; Vandenbussche, 2022). Furthermore, some YDT can begin to experience, after discontinuation, a reverse form of dysphoria developed

as a result of the body changes induced by gender-affirming medical care (Pullen Sansfaçon, Gelly, et al., 2023). These studies may help explaining why providers, in our study, identified the presence of mental health issues among youth in their caseloads. The providers of our study experienced both situations where treatments appeared to have improved youth mental health while others did not. In Littman's study (2021) some of gender-dysphoric people have discontinued gender-affirming treatments because they did not improve their mental health. This highlights the importance of providers working with TNBY to ensure that they are supported in their global mental health needs. Providers should propose a variety of interventions adapted to youth needs, before, during, and after gender-affirming medical care, as youth, regardless of their gender outcomes, may continue to need support to deal with mental health

In our study, over a third of providers described youth as being very well supported by their parents, and half as having parents who took the time to accept and support them in their transition. Very few providers have said to have observed youth experiencing negative or abusive situations from their families. However, just under a third of the trans-care providers stated that some parents seemed reluctant about the transition of their child and had doubts about their youth's trans identity. In terms of social support, most

Table 5 *Providers' Observations of YDT Social Support*

Item	Yes % (n)	A bit % (n)	No % (n)		N = 34
The youth have been well supported by their friends during transition	70.6 (24) ^a	14.7 (5)	2.9(1)	+4	
The youth had access to the medical services they needed	67.6 (23) ^a	14.7 (5)	11.8 (4)	+2	
The youth had access to the psychosocial services they needed	58.8 (20) ^a	29.4 (10)	8.8 (3)	+1	
The youth had access to a supportive community	$50 (17)^{a}$	35.3 (12)	8.8 (3)	+2	
It seems like a number of obstacles were present in the youth's life before transition was discontinued	29.4 (10)	35.3 (12) ^a	32.4 (11)	+1	
The youth reported feeling a certain amount of pressure from their friends about the way in which they were living their gender	5.9 (2)	23.5 (8)	64.7 (22) ^a	+2	
The youth described their school as supportive during their transition	50 (17) ^a	23.5 (8)	8.8 (3)	+6	

Note. Response options: Yes: describes the situation well; A bit: somewhat describes the situation; No: does not describe the situation; N/A: not applicable/I don't know; YDT = young people who have discontinued a transition. ^aMode, the most frequently represented value. + Number of participants who responded N/A.

providers described the YDT as very well supported by their friends. They also reported in greater numbers that YDT were able to benefit from the health and psychosocial services that they needed.

Our results contrast with those of Turban et al. (2021) who identified external factors (i.e., pressure from parents, stigma, discrimination) as very important factors contributing to the experience of discontinuing a transition. Our data shows that, from the perspective of the providers, social pressure may play a role in some situations of discontinuation, but that external factors are less often involved. Given that external factors appear to be particularly present in gender journey that led to temporary discontinuation (Turban et al., 2021), and the importance of parental and social support factors for TNBY's mental health (Tankersley et al., 2021), professional interventions should make concerted effort to take into account the youth's overall environment.

The health providers surveyed also said that YDT they encountered in their caseload did not necessarily express regret. Research shows that TNBY in favorable environments can discontinue without feeling rejection, distress, or regret (Durwood et al., 2022). However, 40% of the providers said that YDT experienced regret, but this result did not provide further details on the type of regret experienced, and only gives little information on its intensity. The feeling of regret can be varied and also be fleeting (Narayan et al., 2021) and some could experience ambivalent feelings about their past experience (MacKinnon et al., 2021; Pullen Sansfaçon, Gelly, et al., 2023). In our research, we have found almost as many situations where providers found youth to be happy to have started and stopped a transition (35.3%), as a situation where they were not happy (38.2%). This potentially reflects that for some, the experience of transition/discontinuation may be positive and could be seen as a healthy part of their evolving identity and creativity (Butler & Hutchinson, 2020; Pullen Sansfaçon, Gelly, et al., 2023). However, unfortunately, others may have experienced what Hildebrand-Chupp (2020) called the negative transition experience, which could also include regret. Providers therefore have an important role to play in accompanying YDT in the various emotions and challenges they may encounter during this period. Because of the heterogeneity of experiences found among YDT (Expósito-Campos, 2021), and the possible presence of regret, trauma and mental health issues, providers should develop interventions that are flexible and nonconstraining, allowing for youth to develop trust in them and to support the exploration of their experience as well as the positive and the negative feelings that may result from discontinuation. In their study of YDT, Pullen Sansfaçon, Gelly, et al., 2023 suggest that ambiguous loss theory (Boss, 2007) may be a useful starting point to help the person finding meaning in their experience of discontinuation and to support the development of resilience, self-acceptance, and growth. Through helping the person to accept that ambivalence is a normal process and that it can emerge in complex experiences such as discontinuation, ambiguous loss theory allows the person to move beyond regret, and to find meaning for their ambiguous feelings, may they be triggered by internal sources (e.g., conflicting thoughts about one's decision to start a transition, experience of gender dysphoria that evolve but that not necessarily go away in some cases) or emerging as a result of external sources or pressure (e.g., pressure of family to conform to certain gender roles, experience of losing friends or family members through transition; having to experiment with new gender roles; Pullen Sansfaçon, Gelly, et al., 2023). As Boss (1999) explains, losses that are left unresolved lead to helplessness. To improve well-being, a person needs to realize that their experience may have left them in a state of ambiguous loss which needs to be accepted. Boss (1999, p. 132) suggests that providers:

Must listen to [the person] as they tell us what their ambiguous losses mean to them, for their stories will vary with culture, gender, race, ethnicity, sexual orientation, and even age. In their narrative will be clues about the source of their distress and thereby the meaning of it: are they upset because they can't figure out what is happening? [...] or are they distressed because they feel helpless and guilty? By listening to their stories, we would gain, not only an understanding of what they are experiencing, but also a real appreciation for their ability to survive and even transcend the pressure thrust upon them by outside forces.

By listening to youth and helping them identifying what is causing ambiguity, providers can engage in an intervention that will help youth finding meaning in the experience, without judgment, and support the development of resiliency and hope. This approach could also possibly be helpful to providers who are working with TNBY who are embarking on a transition process and help them, from the start, to deal with possible ambiguous losses emerging along the way of their gender journey.

Limitations

Despite our large and diverse recruitment, we have a limited sample of providers who completed sufficient sections of the survey and those have been found to mostly use a gender-affirming approach, limiting the generalizability of the data. It is possible that our sampling method (Listserv and social media) introduced a bias in the sample. Also, the data was obtained from providers rather than directly from youth who discontinued their transition, and therefore, must be interpreted with caution. Indeed, providers' observations can be tainted by their interpretation and subjectivity. Moreover, our methodology did not allow us to distinguish the complexities of situations present in professional practices. Even if we noticed a difference in rates between approaches in our data, we did not ask exactly how many youth providers followed beyond 50, which prevents us from proposing a percentage of caseload of discontinuation. The providers did not all have the same work experience and/or do not necessarily follow TNBY regularly in their practice, which possibly limits the possibility of encountering cases of discontinuation. Finally, the study was developed in 2019, and the data collection tools, in early 2020, before many articles on the topic were published. This had an impact on the dimensions that were included in the scales presented in the various tables, as little information on discontinuation was known at that time, leaving the team having to mainly draw on the literature on TNBY and as well as media reports on discontinuation to develop the survey.

Future research should continue to focus on understanding the experiences of trans-health providers and examine, through a larger and more diverse sample of professionals, how intervention approaches may be linked to the likelihood of encountering youth who discontinue their transition. Because it is not possible to predict who will continue or discontinue a transition in the future, research should also examine in more detail, through qualitative research, the practice of trans-care providers who have intervened with youth

beyond the process of transition and of discontinuation in order to understand how youth can be better supported through their entire journey, whatever the outcome of it may be.

Conclusion

From the providers' observation, YDT seems to form a diverse group, a result which is consistent with what can be found in most current studies on the subject. Many providers in our sample have encountered YDT, but most of them only worked with a few, indicating that this phenomenon may not be frequent. Since providers who have encountered YDT in their practice are working in a range of approaches (gender-affirming, watchful waiting, or other approaches), the idea that one specific intervention approach could explain discontinuation among youth is questionable. Therefore, providers should not adopt the stance of trying to avoid discontinuation at any cost by drawing on a particular intervention; rather, they should develop an intervention that supports youth in examining their transition-related needs, that allows for exploration of their hopes, their worries, their questions or their doubts, as well as their capacity to provide consent, and to accompany them during this important journey, whether they will discontinue it or not in the future.

The idea of discontinuation, especially as covered in the media, has already started to limit access to services, while research shows that many TNBY still experience many delays and challenges to access gender-affirming care (Bauer et al., 2021). These restrictions run counter to current knowledge which has shown that access to legal (Cotton et al., 2022), social (Durwood et al., 2017; Russell et al., 2018), and medical (Pullen Sansfaçon, Medico, et al., 2023; Sorbara et al., 2020; Turban et al., 2020) transitions and interventions to the improvement of youth mental health. The findings of this research add to the current knowledge by highlighting how providers have encountered discontinuation as a diverse phenomenon, and are present regardless of their approach to practice.

References

- Ashley, F. (2019a). Gatekeeping hormone replacement therapy for transgender patients is dehumanising. *Journal of Medical Ethics*, 45(7), 480–482. https://doi.org/10.1136/medethics-2018-105293
- Ashley, F. (2019b). Gender (de) transitioning before puberty? A response to Steensma and Cohen-Kettenis (2011). *Archives of Sexual Behavior*, 48(3), 679–680. https://doi.org/10.1007/s10508-018-1328-y
- Ashley, F. (2022). The clinical irrelevance of "desistance" research for transgender and gender creative youth. *Psychology of Sexual Orientation and Gender Diversity*, 9(4), 387–397. https://doi.org/10.1037/sgd0000504
- Baril, A., & Silverman, M. (2019). Forgotten lives: Trans older adults living with dementia at the intersection of cisgenderism, ableism/cogniticism and ageism. Sexualities, 25(6), 1–15. https://doi.org/10.1177/13634 60719876835
- Bauer, G. R., Pacaud, D., Couch, R., Metzger, D. L., Gale, L., Gotovac, S., Mokashi, A., Feder, S., Raiche, J., Speechley, K. N., Temple Newhook, J., Ghosh, S., Pullen Sansfaçon, A., Susset, F., Lawson, M. L., & Trans Youth CAN! Research Team. (2021). Transgender youth referred to clinics for gender-affirming medical care in Canada. *Pediatrics*, 148(5), Article e2020047266. https://doi.org/10.1542/peds.2020-047266
- Boss, P. (1999). Ambiguous loss: Learning to live with unresolved grief. Harvard University Press.
- Boss, P. (2007). Ambiguous loss theory: Challenges for scholars and practitioners [editorial]. Family Relations: An Interdisciplinary Journal of

- Applied Family Studies, 56(2), 105–111. https://doi.org/10.1111/j.1741-3729.2007.00444.x
- Brik, T., Vrouenraets, L. J., de Vries, M. C., & Hannema, S. E. (2020). Trajectories of adolescents treated with gonadotropin-releasing hormone analogues for gender dysphoria. *Archives of Sexual Behavior*, 49(7), 2611–2618. https://doi.org/10.1007/s10508-020-01660-8
- Bustos, V. P., Bustos, S. S., Mascaro, A., Del Corral, G., Forte, A. J., Ciudad, P., Kim, E. A., Langstein, H. N., & Manrique, O. J. (2021). Regret after gender-affirmation surgery: A systematic review and meta-analysis of prevalence. *Plastic and Reconstructive Surgery Global Open*, 9(3), Article e3477. https://doi.org/10.1097/GOX.00000000000003477
- Butler, C., & Hutchinson, A. (2020). Debate: The pressing need for research and services for gender desisters/detransitioners. *Child and Adolescent Mental Health*, 25(1), 45–47. https://doi.org/10.1111/camh.12361
- Cohen, A., Gomez-Lobo, V., Henise, S., Willing, L., Call, D., D'Angelo, L., & Strang, J. (2022). Shifts in gender-related medical requests by transgender and gender diverse adolescents. *Journal of Pediatric and Adolescent Gynecology*, 35(2), Article 233. https://doi.org/10.1016/j.jpag.2022.01 .107
- Cohen-Kettenis, P. T., Schagen, S. E., Steensma, T. D., de Vries, A. L. C., & Delemarre-van de Waal, H. A. (2011). Puberty suppression in a gender-dysphoric adolescent: A 22-year-follow-up. Archives of Sexual Behavior, 40(4), 843–847. https://doi.org/10.1007/s10508-011-9758-9
- Coleman, E., Radix, A. E., Bouman, W. P., Brown, G. R., de Vries, A. L. C., Deutsch, M. B., Ettner, R., Fraser, L., Goodman, M., Green, J., Hancock, A. B., Johnson, T. W., Karasic, D. H., Knudson, G. A., Leibowitz, S. F., Meyer-Bahlburg, H. F. L., Monstrey, S. J., Motmans, J., Nahata, L., ... Arcelus, J. (2022). Standards of care for the health of transgender and gender diverse people, version 8. *International Journal of Transgender Health*, 23(Suppl 1), S1–S259. https://doi.org/10.1080/26895269.2022.2100644
- Cotton, J. C., Martin-Storey, A., Le Corff, Y., Beauchesne Lévesque, S. G., & Pullen Sansfaçon, A. (2022). En Réponse Au Projet De Loi 2: Associations Entre Les Démarches Légales D'affirmation Du Genre et Deux Indicateurs De Bien-être Chez Des Personnes Trans et Non-Binaires Du Québec. *The Canadian Journal of Psychiatry*, 67(7), 578–580. https://doi.org/10.1177/07067437221090088
- Delozier, A. M., Kamody, R. C., Rodgers, S., & Chen, D. (2020). Health disparities in transgender and gender expansive adolescents: A topical review from a minority stress framework. *Journal of Pediatric Psychology*, 45(8), 842–847. https://doi.org/10.1093/jpepsy/jsaa040
- de Vries, A. L. C., & Cohen-Kettenis, P. T. (2012). Clinical management of gender dysphoria in children and adolescents: The Dutch approach. *Journal of Homosexuality*, 59(3), 301–320. https://doi.org/10.1080/ 00918369.2012.653300
- Dhejne, C., Öberg, K., Arver, S., & Landén, M. (2014). An analysis of all applications for sex reassignment surgery in Sweden, 1960–2010: Prevalence, incidence, and regrets. Archives of Sexual Behavior, 43(8), 1535–1545. https://doi.org/10.1007/s10508-014-0300-8
- Drummond, K. D., Bradley, S. J., Peterson-Badali, M., & Zucker, K. J. (2008). A follow-up study of girls with gender identity disorder. Developmental Psychology, 44(1), 34–45. https://doi.org/10.1037/0012-1649.44.1.34
- Durwood, L., Kuvalanka, K. A., Kahn-Samuelson, S., Jordan, A. E., Rubin, J. D., Schnelzer, P., Devor, A. H., & Olson, K. R. (2022). Retransitioning: The experiences of youth who socially transition genders more than once. *International Journal of Transgender Health*, 23(4), 409–427. https://doi.org/10.1080/26895269.2022.2085224
- Durwood, L., McLaughlin, K. A., & Olson, K. R. (2017). Mental health and self-worth in socially transitioned transgender youth. *Journal of the American Academy of Child & Adolescent Psychiatry*, 56(2), 116– 123.e2. https://doi.org/10.1016/j.jaac.2016.10.016
- Edwards-Leeper, L., Leibowitz, S., & Sangganjanavanich, V. F. (2016). Affirmative practice with transgender and gender nonconforming youth:

- Expanding the model. *Psychology of Sexual Orientation and Gender Diversity*, 3(2), 165–172. https://doi.org/10.1037/sgd0000167
- Ehrensaft, D., Giammattei, S. V., Storck, K., Tishelman, A. C., & Keo-Meier, C. (2018). Prepubertal social gender transitions: What we know; what we can learn—A view from a gender affirmative lens. *International Journal of Transgenderism*, 19(2), 251–268. https://doi.org/10.1080/15532739.2017.1414649
- Expósito-Campos, P. (2021). A typology of gender detransition and its implications for healthcare providers. *Journal of Sex & Marital Therapy*, 47(3), 270–280. https://doi.org/10.1080/0092623X.2020.1869126
- Fornander, M. (2022). A mixed-methods examination of transgender youth desistence. Research Days. https://scholarlyexchange.childrensmercy.org/researchdays/GME_Research_Days_2022/ResearchDay1/14
- Gridley, S. J., Crouch, J. M., Evans, Y., Eng, W., Antoon, E., Lyapustina, M., Schimmel-Bristow, A., Woodward, J., Dundon, K., Schaff, R., McCarty, C., Ahrens, K., & Breland, D. J. (2016). Youth and caregiver perspectives on barriers to gender-affirming health care for transgender youth. *Journal of Adolescent Health*, 59(3), 254–261. https://doi.org/10.1016/j.jadohealth 2016/03.017
- Hall, R., Mitchell, L., & Sachdeva, J. (2021). Access to care and frequency of detransition among a cohort discharged by a UK national adult gender identity clinic: Retrospective case-note review. BJPsych Open, 7(6), Article e184. https://doi.org/10.1192/bjo.2021.1022
- Hildebrand-Chupp, R. (2020). More than "canaries in the gender coal mine": A transfeminist approach to research on detransition. *The Sociological Review*, 68(4), 800–816. https://doi.org/10.1177/0038026120934694
- Hunter, J., Butler, C., & Cooper, K. (2021). Gender minority stress in trans and gender diverse adolescents and young people. Clinical Child Psychology and Psychiatry, 26(4), 1182–1195. https://doi.org/10.1177/ 13591045211033187
- Indremo, M., Jodensvi, A. C., Arinell, H., Isaksson, J., & Papadopoulos, F. C. (2022). Association of media coverage on transgender health with referrals to child and adolescent gender identity clinics in Sweden. *JAMA Network Open*, 5(2), Article e2146531. https://doi.org/10.1001/jamanetworkopen .2021.46531
- James, S., Herman, J., Rankin, S., Keisling, M., Mottet, L., & Anafi, M. (2016). The report of the 2015 U.S. Transgender survey. https://ncvc.dspacedirect.org/handle/20.500.11990/1299
- Jones, B. A., Bowe, M., McNamara, N., Guerin, E., & Carter, T. (2021).
 Exploring the mental health experiences of young trans and gender diverse people during the COVID-19 pandemic. *International Journal of Transgender Health*. Advance online publication. https://doi.org/10.1080/26895269.2021.1890301
- Karrington, B. (2022). Defining desistance: Exploring desistance in transgender and gender expansive youth through systematic literature review. *Transgender Health*, 7(3), 189–212. https://doi.org/10.1089/trgh.2020
- Littman, L. (2021). Individuals treated for gender dysphoria with medical and/or surgical transition who subsequently detransitioned: A survey of 100 detransitioners. Archives of Sexual Behavior, 50(8), 3353–3369. https://doi.org/10.1007/s10508-021-02163-w
- MacKinnon, K. R., Ashley, F., Kia, H., Lam, J. S. H., Krakowsky, Y., & Ross, L. E. (2021). Preventing transition "regret": An institutional ethnography of gender-affirming medical care assessment practices in Canada. Social Science & Medicine, 291, Article 114477. https://doi.org/10.1016/j.socscimed.2021.114477
- MacKinnon, K. R., Kia, H., Salway, T., Ashley, F., Lacombe-Duncan, A., Abramovich, A., Enxuga, G., & Ross, L. E. (2022). Health care experiences of patients discontinuing or reversing prior gender-affirming treatments. *JAMA Network Open*, 5(7), Article e2224717. https://doi.org/10.1001/jamanetworkopen.2022.24717
- Marchiano, L. (2017). Outbreak: On transgender teens and psychic epidemics. *Psychological Perspectives*, 60(3), 345–366. https://doi.org/10.1080/00332925.2017.1350804

- Meyer, I. H. (2003). Prejudice, social stress, and mental health in lesbian, gay, and bisexual populations: Conceptual issues and research evidence. Psychological Bulletin, 129(5), 674–697. https://doi.org/10.1037/0033-2909.129.5.674
- Narayan, S. K., Hontscharuk, R., Danker, S., Guerriero, J., Carter, A., Blasdel, G., Bluebond-Langner, R., Ettner, R., Radix, A., Schechter, L., & Berli, J. U. (2021). Guiding the conversation—Types of regret after gender-affirming surgery and their associated etiologies. *Annals of Translational Medicine*, 9(7), Article 605. https://doi.org/10.21037/atm-20-6204
- Olson, K. R., Durwood, L., Horton, R., Gallagher, N. M., & Devor, A. (2022). Gender identity 5 years after social transition. *Pediatrics*, 150(2), Article e2021056082. https://doi.org/10.1542/peds.2021-056082
- Olson-Kennedy, J., Warus, J., Okonta, V., Belzer, M., & Clark, L. F. (2018). Chest reconstruction and chest dysphoria in transmasculine minors and young adults: Comparisons of nonsurgical and postsurgical cohorts. *JAMA Pediatrics*, 172(5), 431–436. https://doi.org/10.1001/jamapediatrics .2017.5440
- Owen-Smith, A. A., Gerth, J., Sineath, R. C., Barzilay, J., Becerra-Culqui, T. A., Getahun, D., Giammattei, S., Hunkeler, E., Lash, T. L., Millman, A., Nash, R., Quinn, V. P., Robinson, B., Roblin, D., Sanchez, T., Silverberg, M. J., Tangpricha, V., Valentine, C., Winter, S.,... Goodman, M. (2018). Association between gender confirmation treatments and perceived gender congruence, body image satisfaction, and mental health in a cohort of transgender individuals. *The Journal of Sexual Medicine*, 15(4), 591–600. https://doi.org/10.1016/j.jsxm.2018.01.017
- Paré-Roy, E., Millette, M., Turbide, O., & Pullen Sansfaçon, A. (2022, September 16–20). De*transition in the media: Transition framed as a mistake; detransition as an undesirable phenomenon. 27th WPATH Symposium, Montreal., QC, Canada.
- Pellicane, M. J., & Ciesla, J. A. (2022). Associations between minority stress, depression, and suicidal ideation and attempts in transgender and gender diverse (TGD) individuals: Systematic review and meta-analysis. Clinical Psychology Review, 91, Article 102113. https://doi.org/10.1016/j.cpr.2021.102113
- Puckett, J., Cleary, P., Rossman, K., Mustanski, B., & Newcomb, M. (2018). Barriers to gender-affirming care for transgender and gender nonconforming individuals. *Sexuality Research and Social Policy*, 15(1), 48–59. https://doi.org/10.1007/s13178-017-0295-8
- Pullen Sansfaçon, A., Gelly, M. A., Gravel, R., Medico, D., Baril, A., Susset, F., & Paradis, A. (2023). A nuanced look into youth journeys of gender transition and detransition. *Infant and Child Development*, 32(2), Article e2402. https://doi.org/10.1002/icd.2402
- Pullen Sansfaçon, A., Medico, D., Riggs, D., Carlile, A., & Suerich-Gulick, F. (2023). Growing up trans in Canada, Switzerland, England, and Australia: Access to and impacts of gender-affirming medical care. *Journal of LGBT Youth*, 20(1), 55–73. https://doi.org/10.1080/19361653.2021.1924918
- Pullen Sansfaçon, A., Temple-Newhook, J., Suerich-Gulick, F., Feder, S., Lawson, M. L., Ducharme, J., Ghosh, S., Holmes, C., ... Stories of Gender-Affirming Care Team. (2019). The experiences of gender diverse and trans children and youth considering and initiating medical interventions in Canadian gender-affirming speciality clinics. *The International Journal of Transgenderism*, 20(4), 371–387. https://doi.org/10.1080/ 15532739.2019.1652129
- Respaut, R., & Terhune, C. (2022). Putting numbers on the rise in children seeking gender care. Reuters Investigate. Retrieved October 6, 2022, from https://www.reuters.com/investigates/special-report/usa-transyouth-data/
- Rood, B. A., Reisner, S. L., Surace, F. I., Puckett, J. A., Maroney, M. R., & Pantalone, D. W. (2016). Expecting rejection: Understanding the minority stress experiences of transgender and gender-nonconforming individuals. *Transgender Health*, 1(1), 151–164. https://doi.org/10.1089/trgh.2016 0012

- Russell, S. T., Pollitt, A. M., Li, G., & Grossman, A. H. (2018). Chosen name use is linked to reduced depressive symptoms, suicidal ideation, and suicidal behavior among transgender youth. *The Journal of Adolescent Health:* Official Publication of the Society for Adolescent Medicine, 63(4), 503– 505. https://doi.org/10.1016/j.jadohealth.2018.02.003
- Slothouber, V. (2020). (De)trans visibility: Moral panic in mainstream media reports on de/retransition. European Journal of English Studies, 24(1), 89– 99. https://doi.org/10.1080/13825577.2020.1730052
- Sorbara, J. C., Chiniara, L. N., Thompson, S., & Palmert, M. R. (2020). Mental health and timing of gender-affirming care. *Pediatrics*, 146(4), Article e20193600. https://doi.org/10.1542/peds.2019-3600
- Spiliadis, A. (2019). Towards a gender exploratory model: Slowing things down, opening things up and exploring identity development. *Metalogos Systemic Therapy Journal*, *35*, 1–9.
- Steensma, T. D., Biemond, R., de Boer, F., & Cohen-Kettenis, P. T. (2011). Desisting and persisting gender dysphoria after childhood: A qualitative follow-up study. *Clinical Child Psychology and Psychiatry*, 16(4), 499– 516. https://doi.org/10.1177/1359104510378303
- Steensma, T. D., McGuire, J. K., Kreukels, B. P. C., Beekman, A. J., & Cohen-Kettenis, P. T. (2013). Factors associated with desistence and persistence of childhood gender dysphoria: A quantitative follow-up study. *Journal of the American Academy of Child & Adolescent Psychiatry*, 52(6), 582–590. https://doi.org/10.1016/j.jaac.2013.03.016
- Tankersley, A. P., Grafsky, E. L., Dike, J., & Jones, R. T. (2021). Risk and resilience factors for mental health among transgender and gender nonconforming (TGNC) youth: A systematic review. Clinical Child and Family Psychology Review, 24(2), 183–206. https://doi.org/10.1007/s10567-021-00344-6
- Temple Newhook, J., Pyne, J., Winters, K., Feder, S., Holmes, C., Tosh, J., Sinnott, M.-L., Jamieson, A., & Pickett, S. (2018). A critical commentary on follow-up studies and "desistance" theories about transgender and gender-nonconforming children. *International Journal of Transgender-ism*, 19(2), 212–224. https://doi.org/10.1080/15532739.2018.1456390
- Tordoff, D. M., Wanta, J. W., Collin, A., Stepney, C., Inwards-Breland, D. J., & Ahrens, K. (2022). Mental health outcomes in transgender and nonbinary youths receiving gender-affirming care. *JAMA Network Open*, 5(2), Article e220978. https://doi.org/10.1001/jamanetworkopen.2022.0978
- Turban, J. L., Carswell, J., & Keuroghlian, A. S. (2018). Understanding pediatric patients who discontinue gender-affirming hormonal interventions. *JAMA Pediatrics*, 172(10), 903–904. https://doi.org/10.1001/jamapediatrics. .2018.1817
- Turban, J. L., & Keuroghlian, A. S. (2018). Dynamic gender presentations: Understanding transition and "de-transition" among transgender youth. *Journal of the American Academy of Child & Adolescent Psychiatry*, 57(7), 451–453. https://doi.org/10.1016/j.jaac.2018.03.016

- Turban, J. L., King, D., Carswell, J. M., & Keuroghlian, A. S. (2020).Pubertal suppression for transgender youth and risk of suicidal ideation.Pediatrics, 145(2), e20191725. https://doi.org/10.1542/peds.2019-1725
- Turban, J. L., Loo, S. S., Almazan, A. N., & Keuroghlian, A. S. (2021). Factors leading to "detransition" among transgender and gender diverse people in the United States: A mixed-methods analysis. *LGBT Health*, 8(4), 273–280. https://doi.org/10.1089/lgbt.2020.0437
- Vandenbussche, E. (2022). Detransition-related needs and support: A cross-sectional online survey. *Journal of Homosexuality*, 69(9), 1602–1620. https://doi.org/10.1080/00918369.2021.1919479
- van Leerdam, T. R., Zajac, J. D., & Cheung, A. S. (2023). The effect of gender-affirming hormones on gender dysphoria, quality of life, and psychological functioning in transgender individuals: A systematic review. *Transgender Health*, 8(1), 6–21. https://doi.org/10.1089/trgh.2020.0094
- Vrouenraets, L. J. J. J., de Vries, M. C., Hein, I. M., Arnoldussen, M., Hannema, S. E., & de Vries, A. L. C. (2022). Perceptions on the function of puberty suppression of transgender adolescents who continued or discontinued treatment, their parents, and clinicians. *International Journal* of Transgender Health, 23(4), 428–441. https://doi.org/10.1080/2689526 9.2021.1974324
- Wallien, M. S., & Cohen-Kettenis, P. T. (2008). Psychosexual outcome of gender-dysphoric children. *Journal of the American Academy of Child* & Adolescent Psychiatry, 47(12), 1413–1423. https://doi.org/10.1097/ CHI.0b013e31818956b9
- Wiepjes, C. M., Nota, N. M., de Blok, C. J., Klaver, M., de Vries, A. L.,
 Wensing-Kruger, S. A., de Jongh, R. T., Bouman, M.-B., Steensma, T.
 D., Cohen-Kettenis, P., Gooren, L., Kreukels, B., & den Heijer, M.
 (2018). The Amsterdam Cohort of Gender Dysphoria Study (1972–2015): Trends in prevalence, treatment, and regrets. *The Journal of Sexual Medicine*, 15(4), 582–590. https://doi.org/10.1016/j.jsxm.2018.01.016
- Winters, K. (2017). Revisiting flawed research behind the 80% childhood gender dysphoria "desistance" myth. https://gidreform.wordpress.com/ 2017/02/10/revisiting-flawed-research-behind-the80-childhood-genderdysphoria-desistance-myth/
- Winters, K., Temple Newhook, J., Pyne, J., Feder, S., Jamieson, A., Holmes, C., Sinnott, M.-L., Pickett, S., & Tosh, J. (2018). Learning to listen to trans and gender diverse children: A response to Zucker (2018) and Steensma and Cohen-Kettenis (2018). *International Journal of Transgenderism*, 19(2), 246–250. https://doi.org/10.1080/15532739.2018.1471767

Received May 2, 2022
Revision received March 20, 2023
Accepted March 21, 2023